EECS 244 Fall 1997, lecture 1.1 Prof . A. Richard Newton

University of California at Berkeley

Page 1 Copyright © 1997, A.Richard Newton

EECS 244

Computer-Aided Design of

Integrated Circuits and Systems

Professor A. Richard Newton

Room 566 Cory Hall

642-2967, rnewton @ ic . eecs

Office Hours: Tu. Th. 3:30-4:30pm

Fall 1997

Adm inistrative Details

m Prof . A . Ri chard Newton

566 Cory Hall , 2-2967, rnewton @ i c . berkel ey . edu

OH: Tu. Th. 3:30-4:30pm

m T.A.: Mi chael Shi l m an , m i chael s @ eecs . berkel ey . edu

m Lecture: Tu. Th. 2-3:30pm , 203 Mc Laughl i n

m Grading:

Hom eworks : 40%

Fi nal Project: 60%

EECS 244 Fall 1997, lecture 1.1 Prof . A. Richard Newton

University of California at Berkeley

Page 2 Copyright © 1997, A.Richard Newton

Goals of the Course

m Introduce f undam ental concepts and al gorithm s used i n the